Author's response to reviews

Title: An unusual case of metastasis to the right ventricular: a case report

Authors:

Guodong Xu (xuguodong5306750@gmail.com)
Guofeng Shao (guofengshaolihuili@163.com)
Shibao Shi (xinpao@hotmail.com)

Version: 2 Date: 26 January 2013

Author's response to reviews:

Dear Editor,

We are sorry for the mistakes made for our ignorance of the instructions for authors. By checking the instructions for authors carefully, we have modified the manuscript. Meanwhile, we replace the terms describing the patient’s gender with the term “female” and added the ethnicity of the patient in the case presentation section of the abstract and the manuscript. Also we replace dates found in the case presentation to the amount of time this occurred before the presentation of the case report.

Thank you very much for your advice on our manuscript. I'm looking forward to hearing from you soon.

Sincerely yours,

Guodong Xu, MD